\begin{table}%t3
\caption{\label{table_p2}Parameters of the considered transiting planetary systems.}
%\centerline
{                \begin{tabular}{lcccccr}
                \hline\hline
                                                 Name & $ v \sin i$ & $P_{\rm rot}$ & Age & $\tau_{\rm synch}$ & $e$ & References \\
                       & (kms$^{-1}$) & (days) & (Gyr) & (Gyr) &  & \\\hline
                         & & & & & & \\ 
        HD149026  &   6.00 $\pm $ 0.50  &   &   2.00- 3.60  &    34.10  &  0.0   &        Wo07 Sa05       \\ 
        HD189733  &            & 11.953 $\pm$ 0.009  &               &           &  0.004 $\pm $  0.003  &  He07 Mo07 Po07 Wi07   \\ 
   HD197286/WASP7  &  17.00 $\pm $ 2.00  &   &               &           &  0.0  &           He09         \\ 
        HD209458  &     & 11.4 $\pm$ 1.5  &               &           &  0.015 $\pm $  0.005  &       Wi05 Wi05a       \\ 
          TrES-1  &   1.30 $\pm $ 0.30  &   &               &           &  0.0   &    Na07 Wi07a Al04     \\ 
          TrES-2  &   1.00 $\pm $ 0.60  &   &   2.80- 7.80  &     0.56  &  0.0   &        Wi08 So07       \\ 
          TrES-3  &   1.50 $\pm $ 1.00  &   &   0.10- 3.70  &     0.04  &  0.0   &           So09         \\ 
          TrES-4  &   8.50 $\pm $ 0.50  &   &   2.50- 4.40  &    10.45  &  0.0  &           So09         \\ 
            XO-1  &   1.11 $\pm $ 0.70  &   &   0.60- 5.50  &     9.31  &  0.0   &        Ho06 Mc06       \\ 
            XO-2  &   1.30 $\pm $ 0.50  &   &               &           &  0.0   &        Fe09 Bu07       \\ 
            XO-4  &   8.80 $\pm $ 0.50  &   &   1.50- 2.70  &    12.03  &  0.0   &           Mc08         \\ 
            XO-5  &   0.70 $\pm $ 0.50  &   &  12.80-16.80  &     1.88  &  0.010 $\pm $  0.013  &           Pa09         \\ 
         HAT-P-1  &   3.75 $\pm $ 0.58  &   &   1.60- 4.60  &   130.20  &  0.0   &   Jo08 Wi07a Ba07a     \\ 
         HAT-P-3  &   0.50 $\pm $ 0.50  &   &   0.10- 6.90  &           &  0.0   &           To07         \\ 
         HAT-P-4  &   5.50 $\pm $ 0.50  &   &   3.60- 6.80  &     7.87  &  0.0   &        To08 Ko07       \\ 
         HAT-P-5  &   2.60 $\pm $ 1.50  &   &   0.80- 4.40  &     2.53  &  0.0   &       To08 Ba07b       \\ 
         HAT-P-6  &   8.70 $\pm $ 1.00  &   &   1.60- 2.80  &    28.17  &  0.0   &        To08 No08       \\ 
         HAT-P-8  &  11.50 $\pm $ 0.50  &   &   2.40- 4.40  &     7.33  &  0.0   &           La08         \\ 
         HAT-P-9  &  11.90 $\pm $ 1.00  &   &   0.20- 3.40  &   204.30  &  0.0   &        Am09 Sh09       \\ 
  HAT-P-10/WASP11  &   0.50 $\pm $ 0.20  &   &   4.10-11.70  &    11.05  &  0.0   &       We09 Ba09a       \\ 
        HAT-P-12  &   0.50 $\pm $ 0.40  &   &   0.50- 4.50  &    38.73  &  0.0   &           Ha09         \\ 
        HAT-P-13  &   4.10 $\pm $ 0.50  &   &   4.20- 7.50  &     2.79  &  0.021 $\pm $  0.009  &          Ba09b         \\ 
          WASP-1  &   5.79 $\pm $ 0.35  &   &   1.00- 3.00  &     3.11  &  0.0   &  Ca07 St07 Sh07 Ch07   \\ 
          WASP-2  &       &       &               &           &  0.0   &  Ca07 St07 Sh07 Ch07   \\ 
          WASP-3  &  13.40 $\pm $ 1.50  &   &               &           &  0.0   &        Gi08 Po08       \\ 
          WASP-4  &   2.00 $\pm $ 1.00  &   &               &           &  0.0   &           Gi09         \\ 
          WASP-5  &   3.50 $\pm $ 1.00  &   &               &           &  0.038 $\pm $  0.026  &           Gi09         \\ 
          WASP-6  &   1.60 $\pm $ 0.30  &   &               &           &  0.054 $\pm $  0.018  &          Gi09a         \\ 
          WASP-8  &       &       &               &           &  0.0   &           Sm09         \\ 
         WASP-10  &     & 11.91 $\pm$ 0.05  &               &           &  0.059 $\pm $  0.014  &        Jo09 Sm09       \\ 
         WASP-12  &   2.20 $\pm $ 1.50  &   &               &           &  0.049 $\pm $  0.015  &           He09         \\ 
         WASP-13  &   2.50 $\pm $ 2.50  &   &   3.60-14.00  &           &  0.0   &           Sk09         \\ 
         WASP-15  &   4.00 $\pm $ 2.00  &   &   2.60- 6.70  &    22.24  &  0.0   &           We09a         \\ 
         WASP-16  &   3.00 $\pm $ 1.00  &   &   0.10- 8.10  &    12.21  &  0.0   &           Li09         \\ 
         WASP-18  &  11.00 $\pm $ 1.50  &   &   0.50- 1.50  &  $6.5\times 10^{-4}$  &  0.009 $\pm $  0.003  &          He09a         \\ 
         CoRoT-2  &    & 4.52 $\pm$ 0.15  &   0.10- 1.70  &     0.36  &  0.0   &        Bo08 Al08       \\ 
         CoRoT-3  &  17.00 $\pm $ 1.00  &   &   1.60- 2.80  &     1.41  &  0.008 $\pm $  0.015  &        De08 Tr09       \\ 
         CoRoT-4  &    & 9.20 $\pm$ 0.3  &   0.70- 2.00  &  2728.00  &  0.0 $\pm $  0.100  &     Ai08 Mo08 La09     \\ 
         CoRoT-5  &   1.00 $\pm $ 1.00  &   &   5.50- 8.50  &           &  0.090 $\pm $  0.090  &           Ra09         \\ 
      OGLE-TR-L9  &  39.33 $\pm $ 0.40  &   &   0.10- 0.70  &     0.92  &  0.0   &           Sn09         \\ 
      OGLE-TR-10  &   7.70 $\pm $ 3.00  &   &               &           &  0.0   & To04 Bo05 Ko05 Po07a   \\ 
      OGLE-TR-56  &   3.00 $\pm $ 3.00  &   &               &           &  0.0   & To04 Bo05 Ko05 Po07a   \\ 
     OGLE-TR-111  &       &       &               &           &  0.0   &        Po04 Mi07       \\ 
     OGLE-TR-113  &   3.00 $\pm $ 3.00  &   &               &           &  0.0  &     Bo04 Di07 Ko04     \\ 
     OGLE-TR-132  &   3.00 $\pm $ 3.00  &   &   0.10- 1.40  &           &  0.0   &     Gi07 Mo04 Bo04     \\ 
     OGLE-TR-182  &       &       &               &           &  0.0   &          Po08a         \\ 
     OGLE-TR-211  &       &       &               &           &  0.0   &           Ud08         \\ 
\hline
                           \end{tabular}}
	       \medskip
   \tablebib{Ai08:          \citet{Aigrainetal08}; 
   Al04:           \citet{Alonsoetal04}; 
   Al08:           \citet{Alonsoetal08}; 
   Am09:        \citet{Ammler-vonetal09};
   Ba07:         \citet{Bakosetal07};  
  Ba07a:           \citet{Bakosetal07a}; 
  Ba07b:           \citet{Bakosetal07b}; 
  Ba09a:           \citet{Bakosetal09a}; 
  Ba09b:           \citet{Bakosetal09c}; 
   Bo04:           \citet{Bouchyetal04}; 
   Bo05:           \citet{Bouchyetal05}; 
   Bo08:           \citet{Bouchyetal08}; 
   Bu07:            \citet{Burkeetal07}; 
   Ca07:          \citet{Cameronetal07}; 
   Ch07:      \citet{Charbonneauetal07}; 
   De08:          \citet{Deleuiletal08}; 
   Di07:             \citet{Diazetal07}; 
   Fe09:        \citet{Fernandezetal09}; 
   Gi07:           \citet{Gillonetal07}; 
   Gi08:           \citet{Gibsonetal08}; 
   Gi09:           \citet{Gillonetal09}; 
  Gi09a:          \citet{Gillonetal09a}; 
   Ha09:          \citet{Hartmanetal09}; 
   He08:            \citet{HenryWinn08}; 
   He09:             \citet{Hebbetal09}; 
  He09a:          \citet{Hellieretal09}; 
  He09b:         \citet{Hellieretal09a}; 
   Ho06:           \citet{Holmanetal06}; 
   Jo08:          \citet{Johnsonetal08}; 
   Jo09:            \citet{Joshietal09}; 
   Ko04:          \citet{Konackietal04}; 
   Ko05:          \citet{Konackietal05}; 
   Ko07:           \citet{Kovacsetal07}; 
   La08:           \citet{Lathametal08}; 
   La09:            \citet{Lanzaetal09b}; 
%   Le09:            \citet{Legeretal09}; 
   Li09:           \citet{Listeretal09}; 
   Mc06:       \citet{McCulloughetal06}; 
   Mc08:       \citet{McCulloughetal08}; 
   Mi07:          \citet{Minnitietal07}; 
   Mo04:           \citet{Moutouetal04}; 
   Mo07:           \citet{Moutouetal07}; 
   Mo08:           \citet{Moutouetal08}; 
   Na07:           \citet{Naritaetal07}; 
   No08:            \citet{Noyesetal08}; 
   Pa09:              \citet{Paletal09}; 
   Po04:             \citet{Pontetal04}; 
   Po07:             \citet{Pontetal07}; 
  Po07a:            \citet{Pontetal07a}; 
   Po08:         \citet{Pollaccoetal08}; 
  Po08a:            \citet{Pontetal08a}; 
%   Qu09:           \citet{Quelozetal09}; 
   Ra09:            \citet{Raueretal09}; 
   Sa05:             \citet{Satoetal05}; 
   Sh07:          \citet{Shporeretal07}; 
   Sh09:          \citet{Shporeretal09}; 
   Sk09:          \citet{Skillenetal09}; 
   Sm09:            \citet{Smithetal09}; 
   Sn09:          \citet{Snellenetal09}; 
   So07:         \citet{Sozzettietal07}; 
   So09:         \citet{Sozzettietal09}; 
   St07:         \citet{Stempelsetal07}; 
   To04:           \citet{Torresetal04}; 
   To07:           \citet{Torresetal07}; 
   To08:           \citet{Torresetal08}; 
   Tr09:           \citet{Triaudetal09}; 
   Ud08:          \citet{Udalskietal08}; 
   We09:            \citet{Westetal09a}; 
   We09a:           \citet{Westetal09b}; 
   Wi05:             \citet{Winnetal05}; 
  Wi05a:       \citet{Wittenmyeretal05}; 
   Wi07:             \citet{Winnetal07}; 
  Wi07a:            \citet{Winnetal07a}; 
  Wi07b:            \citet{Winnetal07b}; 
   Wi08:             \citet{Winnetal08}; 
   Wo07:             \citet{Wolfetal07}. }
                             \end{table}